Author's response to reviews

Title: Determinants of unintended pregnancies in rural Ghana

Authors:

Sebastian Eliason (sakeliason@yahoo.co.uk)
Frank Baiden (baidenf@yahoo.co.uk)
Barbara A Yankey (hellobabs8@yahoo.com)
Kofi Awusabo-Asare (awusabo_asare@yahoo.co.uk)

Version: 4
Date: 13 February 2014

Author's response to reviews:

The Editor-In-Chief
BMC Pregnancy and Childbirth
UK.

MS: 1103988124117735 :Research article: Determinants of unplanned pregnancies in rural Ghana

Dear EIC,

Thanks for the initial review of the above article. The article has been revised based on the editorial comments sent on monday february 10th as follows:

1. The outcome of interest that this article seeks to discuss or address is technically not unplanned pregnancies but rather unintended pregnancies. We in Ghana have over the years used unplanned and unintended pregnancies interchangeably, but that is technically incorrect. The title and contents where applicable have been revised to read 'unintended' instead of 'unplanned'.

2. "Unplanned pregnancies carry serious consequences" - has been replaced by "may carry serious consequences",

3. "Questions asked related to " has been replaced by " Questions related to "

4. " Within the period, 1,914 were interviewed. has been replaced by " Within the study period, 1,914 women were interviewed.

5. "religious sects" has been replaced by "religious groups"

6. Sentence has been revised to read "had significantly lower odds of having unintended pregnancies than........

7. This (0.96, CI 0.94-0.97) and others in the abstarct and text have been revised to read thus (OR 0.96, 95%CI 0.94-0.97)

8. All P values reading 0.000 have been revised to read P<0.001 in tables
Thank you
Dr S. Eliason
Author for Correspondence